Author's response to reviews

Title: The Evaluating Self-Management and Educational Support in Severely Obese Patients Awaiting Multidisciplinary Bariatric Care (EVOLUTION) Trial: Rationale and Design

Authors:

Raj S Padwal (rpadwal@ualberta.ca)
Arya M Sharma (amsharm@ualberta.ca)
Miriam Fradette (miriamf@ualberta.ca)
Susan Jelinski (susan.jelinski@albertahealthservices.ca)
Scott W Klarenbach (swk@ualberta.ca)
Alun Edwards (alun.edwards@albertahealthservices.ca)
Sumit R Majumdar (me2.majumdar@ualberta.ca)

Version: 2 Date: 25 June 2013

Author's response to reviews: see over
Thanks, please see below:

Therefore, could you please clarify whether you have ethical approval and external funding for your study, and if so, could you send evidence of your approvals to the address below?
Yes, they have been sent to you. The study is funded by CIHR. Ethics approval from the U of Alberta REB has been secured.

Could you also provide a list of any manuscripts that are under submission with other journals based on this study protocol? None.

BioMed Central requires all controlled clinical trials to be registered in a suitable publicly accessible registry prior to consideration for publication. The trial registers that currently meet all of the ICMJE guidelines can be found at http://www.icmje.org/faq.html.
Yes, it is indicated in the manuscript – registered at clinicaltrials.gov

**Trial Identifier:** NCT01860131

**EDITORIAL REQUESTS**

1. Competing interests:

Manuscripts should include a ‘Competing interests’ section. This should be placed after the Conclusions/Abbreviations. Please consider the following questions and include a declaration of competing interests in your manuscript:

DONE

Financial competing interests

DONE

Non-financial competing interests

DONE

2. Authors' Contributions

DONE

3. Acknowledgements

DONE
4. Please format your abstract according to our guidelines

(http://www.biomedcentral.com/bmchealthservres/authors/instructions/studyprotocol#formatting-abstract)

DONE

THANKS!
Raj Padwal